Author's response to reviews

Title: Efficacy of manual therapy treatments for people with cervicogenic dizziness and pain: Protocol of a randomised controlled trial

Authors:

Susan A Reid (Sue.Reid@newcastle.edu.au)
Darren A Rivett (Darren.Rivett@newcastle.edu.au)
Michael Katekar (katekar@westnet.com.au)
Robin Callister (Robin.callister@newcastle.edu.au)

Version: 2 Date: 21 July 2012

Author's response to reviews: see over
20th July 2012

Dear Editors,
Thank you for considering our manuscript for publication in the Journal BMC Musculoskeletal Disorders.

1) We have added a statement to the Methods section stating that
The study design and procedures were approved by the University of Newcastle Human Research Ethics Committee (Protocol Number: H-2009-0377), and the procedures followed were in accordance with the Helsinki Declaration of 1975, as revised in 2000.

2) The study protocol has undergone peer review by the Mulligan Concept Teachers Association who awarded a research grant to fund this clinical trial (see attached evidence for both types of approval). The "Mulligan Concept Teachers Association did review the study protocol. We were able to request changes to the protocol if needed. The Mulligan Concept teachers association did provide funding for this study.

Yours sincerely,

Sue Reid